Author's response to reviews

**Title:** Impact of organised programs on colorectal cancer screening

**Authors:**

Francois Eisinger (eisinger@marseille.inserm.fr)
Laurent Cals (laurent.cals@ch-toulon.fr)
Anne Calazel-Benque (acalazel@capio.fr)
Jean-Yves Blay (blay@lyon.fnclcc.fr)
Yvan Coscas (DRYC43@aol.com)
Sylvie Dolbeault (sylvie.dolbeault@curie.net)
Moise Namer (moise.namer@wanadoo.fr)
Xavier Pivot (xavier.pivot@univ-fcomte.fr)
Olivier Rixe (olivier.rixe@psl.ap-hop-paris.fr)
Daniel Serin (d.serin@isc84.org)
Claire Roussel (claire.roussel@roche.com)
Jean-Francois Morere (jean-francois.morere@avc.ap-hop-paris.fr)

**Version:** 3  **Date:** 8 April 2008

**Author's response to reviews:**

The running title, keywords and word count had been removed

The affiliations of each author are quoted using superscript numbers

The funding statement has been removed from the title page and appears now in the Acknowledgements section (were it was already)

The abbreviations appear as a paragraph no more as a list

The authors’ contributions had been written as suggested

There is no more shading in the tables